Author’s response to reviews

Title: EUS in diagnosis of pathologically undiagnosed esophageal tuberculosis

Authors:
Rong Zhu (rongzhuzhurong@126.com)
Yonghua Bai (252546069@qq.com)
Yuankun Zhou (1034077302@qq.com)
Xingguo Fang (fangxingguozy@163.com)
Kui Zhao (Kuizhao95868@MSN.com)
Biguang Tuo (tuobiguang@aliyun.com)
Huichao Wu (18076207034@163.com)

Version: 3 Date: 25 Jul 2020

Author’s response to reviews:

Dear editor and reviewers,

Thank you for your kind considerations on our manuscript entitled "EUS in diagnosis of pathologically undiagnosed esophageal tuberculosis" (BMGE-D-20-00351R2). In this revised version, changes to our manuscript were all highlighted within the document by using red colored text.

Technical Comments:

Reviewer 1:
1. There are no three type of ET (so, this is no classification), there are three type of mode of involvement of ET. This needs to be corrected. (Old comment number 8)
Revise: “In summary, ET is often divided into three types according to the method of infection[3, 5]”
“In summary, there are three types of mode of involvement of ET according to the method of infection[3, 5]” (Discussion, line 180, page 9.)

2. References need to be corrected and formatted correctly - They are not in format- e.g. in some surname is first and in few first name is first (ref no.5,6,8,9)
Revise:

